Author's response to reviews

Title: Onset of Efficacy and Tolerability Following the Initiation Dosing of Long-Acting Paliperidone Palmitate: Post-Hoc Analyses of a Randomized, Double-Blind Clinical Trial

Authors:

Cynthia A Bossie (cbossie@its.jnj.com)
Jennifer K Sliwa (jkernsli@its.jnj.com)
Yi-Wen Ma (yma@its.jnj.com)
Dong-Jing Fu (dfu@its.jnj.com)
Larry Alphs (lalphs@its.jnj.com)

Version: 4 Date: 29 March 2011

Author's response to reviews:

Have added the Trial Registration information to the bottom of the abstract as per email from Angelina Ilievska (received on 3-29-2011 and copied below)

Dear Dr Bossie,

Peer review of your manuscript (above) is now complete, and we are delighted, in principle, to accept the manuscript for publication in BMC Psychiatry. However, before pre-acceptance, please include your Trial registration number at the foot of the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Please note that there should be no space between the letters and numbers of the trial registration number.

We look forward to receiving your revised manuscript. If you imagine that it will take longer to prepare please give us some estimate of when we can expect it.

Please don't hesitate to contact me if you have any problems or questions regarding your manuscript.

With best wishes,

Miss Angelina Ilievska MSc
on behalf of Dr Deanna L. Kelly
Tel: +44 (0) 20 3192 2013
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/